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ABSTRACT 

Emerging contaminants (ECs) are the latest class of environmental contaminants 

which are highly recalcitrant, and toxic and currently lack specific regulatory guidelines. In 

the absence of suitable treatment methods against ECs, their concentration is increasing and 

has reached a level impacting human health and ecology. Therefore, to mitigate the effect of 

ECs, suitable treatment methods are required, and recently, plasma technology has shown 

potential for ECs degradation in a sustainable manner. Therefore, this work was conducted to 

study the suitability of Dielectric Barrier Discharge (DBD) plasma against ECs treatment 

from real wastewater (WW) using three different plasma reactors. Initially, the DBD plasma 

feasibility study was performed in a single jet plasma system, and degradation of dyes 

Rhodamine B (Rd. B) and Methylene Blue (Met. Blue) was performed. The DBD plasma jet 

provided a complete degradation and a 70% mineralization of both dyes in 30 min. The 

plasma jet efficiency was then evaluated for pharmaceuticals degradation in real WW, and 

90% degradation efficiency was obtained, showing plasma's effectiveness in real 

environmental conditions. Then, to evaluate the DBD plasma jet's suitability with the 

conventional WW treatment system, a hybrid system consisting of the sequencing batch 

reactor (SBR) and DBD plasma jet was developed. The hybrid system with complete removal 

of the pharmaceuticals also provided a 50% mineralization, showing the plasma system's 

feasibility for real environmental applications. Continuing this, an attempt was made to scale 

up the plasma jet system, and a multiarray plasma jet system was developed, but the inter-jet 

interactions and high feed gas flow rate (>30 ml/min) requirement affected the uniform 

plasma formation. Therefore, improving the plasma systems capacity in the form of MultiJet 

was considered unsuitable, especially for its application in WW treatment. The plasma jet 

study showed that the limited plasma area is a major limitation for their utilization for 

practical purposes, and to overcome this issue, a novel S-shaped neon DBD plasma reactor 



v 
 

was designed in which curved electrodes were used in a linear configuration, providing a flat 

filamented plasma spreading entirely over the electrode length and the liquid surface. A low-

cost commercial neon power supply (12 kV and 30 mA) was used, and the plasma generation 

capacity was compared with air, argon, and oxygen as the feed gases. The larger plasma area 

provided by the neon plasma system enhanced the plasma-liquid interface, thereby improving 

the diffusion of reactive species into the bulk liquid. This increased diffusion of reactive 

species was anticipated to increase the degradation of contaminants present in the bulk liquid. 

However, the removal of surface-active compounds exceeded that of the bulk contaminants 

due to the preferential interactions between the surface-active compounds and the reactive 

species at the plasma-liquid interface. To better understand the role of surface activity on 

contaminants degradation, a comparative study for ECs degradation in neon plasma and 

bubble column plasma reactor (BCPR) was performed. BCPR utilizes the gas bubbles for 

plasma generation, and the continuous sparging of the gas ensures a high homogenization in 

the system. The contaminant's degradation in BCPR provided a linear correlation between the 

contaminant's degradation rate (kobs) and their surface activity (Log KOW) with a preferential 

degradation for more surface-active contaminants. The BCPR was more efficient for 

contaminants degradation, but when it was operated in real environment conditions using 

samples with high initial conductivity and the presence of additional contaminants 

(surfactants and organic pollutants), its degradation efficiency reduced, which showed the 

limitation of BCPR for real WW treatment. No such issue was present with neon plasma, and 

it was also compatible with the conventional wastewater treatment system. Therefore, the 

neon plasma system was found to be more suitable for practical application. Finally, to 

further validate the neon plasma reactor as an effective advanced WW treatment system, a 

risk assessment of plasma-treated samples and an economic assessment of the plasma system 

study were performed, and its results also favored neon plasma.  
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साराांश 

 

उभरते हुए प्रदूषक (ECs) पर्यावरणीर् प्रदूषक ों की एक नवीनतम शे्रणी हैं ज  अत्यधिक अवक्षरण-

प्रधतर िी और धवषैले ह ते हैं, और वतामयन में उनके खिलयफ क ई धवधिष्ट धनर्यमक धदिय-धनदेि नही ों हैं। उपरु्क्त 

उपचयर धवधिर् ों की अनुपखथिधत में, ECs की सयोंद्रतय बढ़ रही है और र्ह मयनव स्वयस्थ्य और पयररखथिधतकी पर 

प्रभयव डयल रही है। इसधलए, ECs के प्रभयव क  कम करने के धलए उपरु्क्त उपचयर धवधिर् ों की आवश्यकतय है, 

और हयल ही में, प्लयज़्मय तकनीक ने ECs के अपघटन के धलए सोंभयधवत समयियन धदियर्य है। इसधलए, इस कयर्ा 

कय उदे्दश्य वयस्तधवक अपधिष्ट जल (WW) से ECs के उपचयर के धलए तीन अलग-अलग प्लयज़्मय ररएक्टर ों कय 

उपर् ग करके डयईलेखक्टि क बैररर्र धडथचयजा (DBD) प्लयज़्मय की उपरु्क्ततय कय अध्यर्न करनय िय। प्रयरोंभ में, 

DBD प्लयज़्मय कय व्यवहयर्ातय अध्यर्न एकल जेट प्लयज़्मय प्रणयली में धकर्य गर्य, और र डयमयइन बी (Rd. B) और 

मेधिलीन बू्ल (Met. Blue) रोंग ों कय अपघटन धकर्य गर्य। DBD प्लयज़्मय जेट ने 30 धमनट में द न ों रोंग ों कय पूरय 

अपघटन और 70% िधनजकरण प्रदयन धकर्य। धफर, वयस्तधवक WW में फयमयासू्यधटकल्स के अपघटन के धलए 

प्लयज़्मय जेट की दक्षतय कय मूल्योंकन धकर्य गर्य, और 90% अपघटन दक्षतय प्रयप्त की गई, ज  वयस्तधवक 

पर्यावरणीर् पररखथिधतर् ों में प्लयज़्मय की प्रभयविीलतय क  दियातय है। धफर, DBD प्लयज़्मय जेट की उपरु्क्ततय कय 

मूल्योंकन करने के धलए, एक हयइधिड प्रणयली धजसमें अनुक्रधमक बैच ररएक्टर (SBR) और DBD प्लयज़्मय जेट 

ियधमल िे, धवकधसत की गई। हयइधिड प्रणयली ने फयमयासू्यधटकल्स क  पूरी तरह से हटयते हुए 50% िधनजकरण भी 

प्रदयन धकर्य, ज  वयस्तधवक पर्यावरणीर् अनुप्रर् ग ों के धलए प्लयज़्मय प्रणयली की व्यवहयर्ातय क  दियातय है। इसके 

सयि ही, प्लयज़्मय जेट प्रणयली क  से्कल-अप करने कय प्रर्यस धकर्य गर्य, और एक मल्टीएरे प्लयज़्मय जेट प्रणयली 

धवकधसत की गई, लेधकन इोंटर-जेट इोंटरैक्शन और उच्च फीड गैस प्रवयह दर (>30 धम.ली./धमनट) की आवश्यकतय 

ने समयन प्लयज़्मय धनमयाण क  प्रभयधवत धकर्य। इसधलए, मल्टीजेट के रूप में प्लयज़्मय धसस्टम की क्षमतय में सुियर 

करनय, धविेष रूप से WW उपचयर में इसके अनुप्रर् ग के धलए अनुपरु्क्त मयनय गर्य। प्लयज़्मय जेट अध्यर्न ने 

धदियर्य धक सीधमत प्लयज़्मय के्षत्र उनके व्ययवहयररक उदे्दश्य ों के धलए उपर् धगतय में एक प्रमुि सीमय है, और इस 

समस्यय क  दूर करने के धलए, एक नवीन S-आकयर की नीर्न DBD प्लयज़्मय ररएक्टर धडजयइन की गई धजसमें 
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वधक्रत इलेक्टि  ड कय उपर् ग धकर्य गर्य, ज  एक रेिीर् कॉखफ़िगरेिन में पूरे इलेक्टि  ड की लोंबयई और तरल सतह 

पर एक समयन प्लयज़्मय फैलयव प्रदयन करतय है। एक कम लयगत वयली वयधणखिक नीर्न धबजली आपूधता (12 kV 

और 30 mA) कय उपर् ग धकर्य गर्य, और प्लयज़्मय उत्पयदन क्षमतय की तुलनय हवय, आगान, और ऑक्सीजन के 

सयि फीड गैस के रूप में की गई। नीर्न प्लयज़्मय प्रणयली द्वयरय प्रदयन धकर्य गर्य बडय प्लयज़्मय के्षत्र प्लयज़्मय-तरल 

इोंटऱेिस क  बढ़यतय है, धजससे प्रधतधक्रर्य प्रजयधतर् ों कय ि क तरल में प्रवयह सुियरतय है। इस प्रकयर, प्रधतधक्रर्यिील 

प्रजयधतर् ों के बढे़ हुए प्रवयह से ि क तरल में उपखथित प्रदूषक ों के अपघटन में वृखि ह ने की सोंभयवनय िी। हयलयोंधक, 

प्लयज़्मय-तरल इोंटऱेिस पर सतही सधक्रर् र्ौधगक ों और प्रधतधक्रर्यिील प्रजयधतर् ों के बीच वरीर्तय इोंटरैक्शन के 

कयरण सतही सधक्रर् र्ौधगक ों कय अपघटन ि क प्रदूषक ों से अधिक ह  गर्य। सतही गधतधवधि के प्रभयव क  बेहतर 

ढोंग से समझने के धलए, नीर्न प्लयज़्मय और बबल कॉलम प्लयज़्मय ररएक्टर (BCPR) में ECs के अपघटन के धलए 

एक तुलनयत्मक अध्यर्न धकर्य गर्य। BCPR गैस बुलबुल ों कय उपर् ग प्लयज़्मय उत्पयदन के धलए करतय है, और गैस 

की धनरोंतर स्पयधजिंग प्रणयली में उच्च समरूपतय सुधनधित करती है। BCPR में प्रदूषक ों के अपघटन ने प्रदूषक ों के 

अपघटन दर (kobs) और उनकी सतही गधतधवधि (Log KOW) के बीच एक रेिीर् सोंबोंि प्रदयन धकर्य, धजसमें अधिक 

सतही सधक्रर् प्रदूषक ों के धलए वरीर्तय अपघटन देिय गर्य। BCPR प्रदूषक ों के अपघटन के धलए अधिक कुिल 

िय, लेधकन जब इसे उच्च प्रयरों धभक धवद्रु्त चयलकतय और अधतररक्त प्रदूषक ों (सफैक्टेंट्स और कयबाधनक प्रदूषक ों) 

की उपखथिधत वयले नमून ों कय उपर् ग करके वयस्तधवक पर्यावरणीर् पररखथिधतर् ों में सोंचयधलत धकर्य गर्य, त  इसकी 

अपघटन दक्षतय कम ह  गई, धजसने वयस्तधवक WW उपचयर के धलए BCPR की सीमय क  धदियर्य। नीर्न प्लयज़्मय 

में ऐसी क ई समस्यय नही ों िी, और र्ह पयरोंपररक अपधिष्ट जल उपचयर प्रणयली के सयि भी सोंगत िय। इसधलए, 

व्ययवहयररक अनुप्रर् ग के धलए नीर्न प्लयज़्मय प्रणयली क  अधिक उपरु्क्त पयर्य गर्य। अोंत में, नीर्न प्लयज़्मय 

ररएक्टर क  एक प्रभयवी उन्नत WW उपचयर प्रणयली के रूप में आगे मयन्य करने के धलए, प्लयज़्मय उपचयररत नमून ों 

की ज खिम मूल्योंकन और प्लयज़्मय प्रणयली अध्यर्न की आधिाक मूल्योंकन की गई, और इसके पररणयम भी नीर्न 

प्लयज़्मय के पक्ष में िे।  
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